Author's response to reviews

Title: Conservation of Resources Theory and Research Use in Health Systems

Authors:

Celeste Alvaro (Celeste.Alvaro@dal.ca)
Renée F Lyons (RLyons@bridgepointhealth.ca)
Grace Warner (Grace.Warner@dal.ca)
Stevan E Hobfoll (Stevan.Hobfoll@rush.edu)
Patricia J Martens (Pat_Martens@cpe.umanitoba.ca)
Ronald Labonté (rlabonte@uottawa.ca)
E. Richard Brown (erbrown@ucla.edu)

Version: 5 Date: 23 September 2010

Author's response to reviews:

Please find attached the latest revised draft of the copy-edited manuscript.
Note that only two modifications remain and I am hoping that you could assist me with the proper citations for software (NVivo 7 and SPSS 15) - these are references numbered 128 and 129.